

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre

Dasgupta, Kabir; Ghimire, Keshar M.; Pacheco, Gail

Working Paper

The impact of smoking during pregnancy on children's body weight

Economics Working Paper Series, No. 2018/04

Provided in Cooperation with:

Faculty of Business, Economics and Law, Auckland University of Technology (AUT)

Suggested Citation: Dasgupta, Kabir; Ghimire, Keshar M.; Pacheco, Gail (2018): The impact of smoking during pregnancy on children's body weight, Economics Working Paper Series, No. 2018/04, Auckland University of Technology (AUT), Faculty of Business, Economics and Law, Auckland

This Version is available at: https://hdl.handle.net/10419/242560

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.





Economics Working Paper Series

Faculty of Business, Economics and Law, AUT

Maternal smoking during pregnancy and child weight outcomes: New evidence from longitudinal data

Kabir Dasgupta

Keshar M. Ghimire

Gail Pacheco

2018/04

Maternal smoking during pregnancy and child weight outcomes: New evidence from longitudinal data

August 2019 Kabir Dasgupta[†], Keshar M. Ghimire[✓], and Gail Pacheco[†]

Abstract

This study explores the impact of mother's smoking during pregnancy on child bodyweight outcomes, from birth through age five. Prior literature suggests that while maternal smoking during pregnancy leads to a decline in birthweight, the same is linked with higher risk of obesity during early childhood years. To examine the causal interpretation of this health phenomenon, we use a nationally representative sample of children and mothers from the National Longitudinal Surveys and exploit exogenous variation in smoking behavior of mothers prompted by changes in federal and state-level tobacco tax rates at the time of conception. Consistent with prior literature, our instrumental variable estimates suggest that children of smokers weigh significantly less at birth than children of non-smokers (represented by an estimated decline of 0.53 kilograms). However, we do not find any evidence that these children are more likely to be overweight during early childhood. In fact, we find some evidence that negative effects of smoking on children's body weight may linger through preschool years.

Keywords: Maternal smoking; pregnancy; birthweight; overweight; instrumental variable; fixed effect.

JEL codes: I10, I12, I18

Correspondence to: New Zealand Work Research Institute, Auckland University of Technology, Private Bag 92006, Auckland 1142, New Zealand. *E-mail address*: kabir.dasgupta@aut.ac.nz

Acknowledgement: We are thankful to the participants and discussants of Southern Economic Association (2017), Tampa USA; American Society of Health Economists (2018), Atlanta USA; New Zealand Association of Economists Conference, Auckland (2018) for providing helpful comments and feedback on the analysis. We are solely responsible for all errors committed in this study.

_

[†] Faculty of Business, Economics & Law, Auckland University of Technology, New Zealand

Department of Economics, University of Cincinnati – Blue Ash, United States of America

1. INTRODUCTION

The effects of maternal smoking during pregnancy on children's bodyweight outcomes, at birth and at later ages, have been widely explored across several academic disciplines, ranging from epidemiology to economics, medicine, and sociology. Broadly, two main conclusions emerge from the literature. First, there is credible evidence of the negative impact of maternal smoking during pregnancy on children's birthweight (Rosenzweig & Wolpin, 1991; Evans & Ringel, 1999; Bharadwaj et al., 2014). Secondly, the weight disadvantage observed at birth disappears at later childhood ages and smokers' children are, in fact, found to be more likely to be overweight compared to non-smokers' children (e.g. Wideroe et al., 2003).

In our context, it is worth noting that existing evidence on the latter (i.e. the link between maternal smoking and childhood obesity) lacks causal interpretation, given the nature of empirical specifications that have been employed in the related literature. To our understanding, this is broadly due to two reasons. First, the empirical specifications commonly adopted in past literature often fail to account for relevant unobserved confounders that can affect maternal behavior and child well-being. This has rendered a majority of studies descriptive in nature, providing correlational rather than causal evidence (more specifics provided in Section 2). Second, the estimates of the effect of smoking on birthweight and at later ages often comes from cross-sectional samples. To combat both concerns raised in prior evidence, this study exploits instrumental variable (IV) estimation strategy and uses a sample of children born to a nationally representative group of American women. The IV strategy in particular aims to provide causal evidence regarding the relationship between maternal smoking during pregnancy and children's bodyweight from birth through age five.

By utilizing matched mother-child data from the National Longitudinal Survey of Youth (NLSY79) and Child and Young Adult Survey (NLS-CYA), our study presents US-based

evidence on the evolutionary impact of maternal smoking during pregnancy on children's bodyweight outcomes from birth through pre-school years. Our empirical strategy is to model children's bodyweight outcomes as a function of their own characteristics, mother's characteristics (including mother's smoking behavior), and family information. One econometric concern in estimating the effect of mothers' smoking on child weight in this framework is the potential endogeneity of mothers' smoking behavior. To address this concern and to estimate the causal relationship of interest, we rely on exogenous variation in maternal smoking participation during pregnancy triggered by Federal and state-level variation in tobacco taxes at the time of conception. In particular, we instrument our key regressor (smoking during pregnancy) with the sum of Federal and State excise tax rates on cigarette consumption (Evans & Ringel, 1999; 2001; Simon, 2016). Our instrument passes the standard tests for strength and validity.

Our contribution is two-fold. First, we confirm the negative effects of maternal smoking during pregnancy on child birthweight documented in prior literature (Rosenzweig & Wolpin, 1991; Evans & Ringel, 1999; Bharadwaj et al., 2014). Second, and more importantly, using the same child-mother sample, we study the evolution of the effects of maternal smoking on children's bodyweight through the pre-school years. In exploring the longer-term effects of maternal smoking during pregnancy, we test the so-called 'catch-up' hypothesis, which postulates that there is an increase in obesity risk among pre-school children who are exposed to prenatal maternal smoking (see Ong et al., 2000; Von Kries et al., 2002).

In line with the findings in the previous literature, our estimates suggest a significant reduction in children's birthweight because of maternal smoking during pregnancy. Specifically, we find

¹ These studies suggest that children of mothers who smoked during pregnancy rapidly grow during early childhood to catch-up with the children of nonsmokers and are in fact more likely to be overweight or obese in later childhood. Ong et al. (2000) focus on likelihood of obesity at the age five whereas Von Kries et al. (2002) focus on the likelihood of being overweight or obese during the age of five to seven years.

that, on average, children of mothers who smoked during pregnancy weigh 0.53 kilogram less than children of mothers who did not smoke while pregnant. However, contrary to what is suggested in the literature, we do not find evidence showing children of smokers are more likely to be overweight in their early childhood years. Instead, in our 2-SLS regression models, we find some evidence for lingering negative effects of maternal smoking during pregnancy on body weight during the pre-school ages.

Our findings, supported by alternative specifications for robustness purposes, have important implications for public health policy. Specifically, our results support and encourage public policies targeted at curbing smoking among expecting mothers and question the causal nature of prior empirical evidence that found a positive relationship between maternal smoking during pregnancy and risk of being overweight or obese in early childhood years. Consequently, it is clear that this is an area with substantial scope for future empirical research.

The remainder of this paper is organized as follows: Section 2 discusses the relevant literature, Section 3 describes the data and variables used in the empirical analysis; Section 4 explains the key elements of the identification strategy; while Section 5 discusses results; and Section 6 presents concluding remarks.

2. RELATED LITERATURE

This paper is related to an extensive literature on the impact of mother's smoking on child health outcomes. Existing studies have linked maternal smoking during pregnancy with a wide array of both short-term and long-term child health consequences. Examples of short-term consequences include premature childbirth, fetal growth restriction, lower birthweight, and infant mortality (Comstock et al., 1971; Meyer & Tonascia, 1977; Cnattingius, 2004). The longer-term health implications for children include higher blood pressure levels, and respiratory and pulmonary disorders (Hanrahan et al., 1992; Stick et al., 1996; Blake et al.,

2000; Li et al., 2016); psychological and behavioral problems - such as attention deficit hyperactivity disorder, neurological problems, and poor cognitive functioning (Weitzman, Gortmaker, & Sobol, 1992; Milberger et al., 1996; Wakschlag et al., 1997; Thapar et al., 2003; Huizink & Mulder, 2006; Key et al., 2007; Gilman, Gardener, & Buka, 2008); and increased risks of childhood obesity (Vik et al.,1996; Fried, Watkinson, & Gray, 1999; Von Kries et al., 2002; Oken, Levitan, & Gillman, 2008).

More closely related to our paper are the studies that specifically evaluate the effect of mothers' smoking during pregnancy on child weight outcomes. Among this set of studies, there are two strands of work to draw on.

The first is well established and implies a strong inverse relationship between children's inutero exposure to maternal smoking and birthweight (Sexton & Hebel, 1984; Permutt & Hebel,
1989; Rosenzweig & Wolpin, 1991; Evans & Ringel, 1999; Harris et al., 2015). A majority of
studies in this space exploit exogenous variation in maternal smoking behavior during
pregnancy resulting from natural experiments such as smoking regulations (e.g. bans or excise
tax on cigarette prices) or controlled trials in an experimental setting where pregnant smokers
are randomly assigned to treatment (smoking cessation interventions such as counselling) and
control groups.

The second strand of relevant literature is focused on child weight outcomes post-birth and argues that the risk of obesity during childhood, mainly during the pre-school period, is much higher among children whose mothers smoke during pregnancy.² The existing evidence in this space is mostly descriptive and thus the associations cannot be interpreted as causal. Our contribution in this literature space is to provide an empirically robust estimate of the causal

² See Oken et al. (2008) and Ino (2010) for a comprehensive review of international studies exploring the association between maternal smoking and child obesity.

impact of mothers' smoking during pregnancy on child weight during pre-school ages (two through five). In following two sub-sections, we discuss the two sets of studies, the methods used, and the key findings in greater detail.

2.1 Causal link between maternal smoking during pregnancy and birthweight

For causal interpretation of the estimated relationship between maternal smoking during pregnancy and children's birthweight, the empirical literature has primarily relied on exogenous variation in smoking behavior prompted by natural experiments such as macro-level regulations implemented to reduce smoking (Evans & Ringel, 1999;2001; Lien & Evans, 2005; Kabir et al., 2009; Bharadwaj et al., 2014). For instance, Kabir et al. (2009; Irish evidence) and Bharadwaj et al. (2014; Norwegian evidence) estimate differences in birthweights before and after implementation of smoking bans in the workplace. Both these studies find that smoking bans were effective in improving (increasing) children's birthweight. With respect to U.S. evidence on this front, Lien & Evans (2005) focus on four states (Arizona, Illinois, Michigan, and Massachusetts) that have experienced large excise tax hikes. Importantly, and of relevance to our study, the authors use state-level regulations as the instrumental variable for smoking participation of mothers during pregnancy. Using two-stage least squares (2-SLS) regressions, the Lien & Evans (2005) find that exogenous variation in maternal smoking during pregnancy has an inverse relationship with children's birthweight. In particular, the authors found that smoking during pregnancy leads to a substantial increase in the likelihood of infants having low birthweight (defined as less than 2500 grams or 5.5 pounds approximately).

The use of variation in excise tax regulation on cigarettes was motivated by earlier work of Evans & Ringel (1999; 2001). More specifically, the authors' 1999 study uses federal and state-level excise tax rates on cigarette consumption to perform 2-SLS regressions to estimate the impact of maternal smoking during pregnancy on children's birthweight. Of note is that the

authors find their IV regression estimate (of a 0.36 kilogram drop in birthweight when there was maternal smoking during pregnancy) to be closely comparable to the results obtained from a clinical trial where pregnant smokers were randomly assigned to an intervention designed to reduce smoking prior to childbirth. Of further note, the 2-SLS estimates increase in magnitude with inclusion of additional socio-demographic characteristics (see Models 1-3 in Table 3 of Evans & Ringel, 1999) and the absolute values are larger than that of the coefficients obtained from single equation models.

The negative correlation between individuals' smoking behavior and cigarette prices (usually increased through taxation) has been widely examined in the previous literature (Wasserman, 1991; Chaloupka & Werner, 2000; Carpenter & Cook, 2008). In majority of empirical studies on the link between smoking and cigarette prices, the estimated price elasticity of demand for cigarettes is found to be in the range between -0.3 to -0.5 (Chaloupka & Werner, 2000). However, these estimates vary by data sources and estimation methods (Evans & Ringel, 1999).

The large evidence on the negative price elasticity of demand for cigarettes prompts us to expect that women are likely to face substantial health as well as economic cost for smoking when pregnant. Therefore, increase in cigarette prices induced through adoption of higher tax rates can provide added impetus to pregnant smokers to reduce smoking behavior in response.

The validity of using governmental smoking regulations as the instrumental variable in this setting is based on the assumption that exogenously determined macro-level policies are uncorrelated with individual-level unobserved heterogeneities that affect maternal behavior and children's well-being at birth. If this is true, then the policy-induced natural experiments can also be utilized to study the effects of maternal smoking during pregnancy on later childhood outcomes as well. Further motivation for using state-level variation in tobacco tax

rates as IV for evaluating the effects of prenatal smoking among mothers on children's long-term outcomes is provided by a recent study by Simon (2016). The study finds that exogenous increase in the state-level cigarette tax rates during pregnancy improves children's long-term health measured in terms of sick leaves from school and incidence of asthma, doctor visits, and hospitalizations.

To conclude this strand of literature review, it is worth mentioning that some studies in this space have also employed alternative empirical strategies such as Fixed Effects (FE) regressions and have found similar results. For example, Rosenzweig & Wolpin (1991), exploit within-mother (across childbirths) variation in prenatal smoking behavior to control for confounding influences arising from mothers' time-invariant characteristics that may affect their prenatal behavior (including smoking) as well as their children's well-being. The authors' findings support the negative relationship between maternal smoking during pregnancy and children's birthweight.

2.2 Association between maternal smoking during pregnancy and early childhood weight

Upon close examination of the extant literature, while there's an overwhelming evidence that prenatal exposure to maternal smoking results in fetal growth retardation (Miller et al., 1976; Cnattingius, 2004; Ward et al., 2007), the same is associated with stunted growth (in terms of height), higher risk of adiposity (skin thickness), and obesogenic growth during later childhood (Oken et al., 2008; Ino, 2010; Suzuki et al., 2011; Howe et al., 2012; Li et al., 2016). Taken together, the literature on maternal smoking during pregnancy indicates that the evolving effects likely prompt smokers' children to experience a greater risk of being overweight (or obese) as they grow up.

Focusing on existing US-based evidence, Whitaker (2004) and Salsberry & Reagan (2007) present the earliest evidence on the positive association between maternal smoking during

pregnancy and an increased risk of being overweight among children. While Whitaker's (2004) study focused at children aged 2-4, Salsberry & Reagan's (2007) population of interest included children aged 2-8. In both studies, the indicator for whether a mother smoked during pregnancy was treated as one of the prenatal determinants of children's bodyweight outcomes rather than the regressor of interest. Similar to our study, Salsberry & Reagan (2007) use linked matched mother-child information from the National Longitudinal Surveys (up to the survey year 2002) to explore the prenatal determinants of risks of having excess weight among children. Estimating multivariate logistic regressions, the authors find smoking during pregnancy (among a range of other prenatal and demographic characteristics) to have a positive and statistically significant association with the likelihood of child being overweight.

Subsequent analysis that have confirmed Salsberry & Reagan's (2007) findings for the U.S., include Chen et al. (2005) and Oken et al. (2005). In these studies, the relevant odds ratio estimated, which represents the risk of being overweight for children whose mothers smoked during pregnancy (compared to non-smokers' children), ranges between 1.21 and 2.20 (Ino, 2010).

With respect to the international evidence, the majority of relevant studies present evidence from developed economies in Europe (e.g. Germany, Sweden, Norway, and Spain) and Australia. Table 1 summarizes some of the empirical findings from selected key studies in the extant literature.

Table 1

Summary of findings from international literature on the association between maternal smoking during pregnancy and risk of being overweight during childhood

Authors	Country (Data period)	Population of interest	1	
Maessen et al. (2019)	Sweden (children born in 1991-2009)	Adult daughters aged 18-26.	Generalized linear model regression	ARR:1.48
Wideroe et al. (2003)	Sweden; Norway (pregnancies in 1986-1992)	Children aged 5	Logistic regression	RR: 2.5
Von Kries et al. (2002)	Germany (school entry data in 1999-2000)	Children aged 5-7	Logistic regression	AOR: 1.43 for overweight; 2.06 for obesity
Mendez et al. (2008)	Spain (children born in 1997-1998)	Children aged 5-7	Logistic regression	AOR (smoking in first trimester): 2.65 AOR (smoking later in pregnancy): 1.88
Al Mamun et. al (2006)	Australia (children born in 1981-1984)	Children aged 14	Logistic regression	OR: 1.31 OR for obesity: 1.42

Notes: OR: Odds ratio; AOR: Adjusted odds ratio; RR: Relative risk ratio; ARR: Adjusted relative risk ratio

It is important to note, that most of the studies discussed thus far attempt to control for 'confounding' factors by including a wide range of observed family-level characteristics as covariates in their regression models such as mother's prenatal attributes and family's socioeconomic status. However, the estimated relationships between smoking during pregnancy and children's likelihood of being overweight may still suffer potential endogeneity bias for not accounting for unobserved heterogeneities that can be related to both maternal behavior and child outcomes. For instance in a recent paper, Bharadwaj et al. (2018) conceptualize how parents respond to children's birth outcomes. The authors demonstrate the possibility that parents may engage in 'compensating behavior' by reallocating parental investments to children born with poorer birth outcomes (such as low birth weight) to mitigate potential long-term differences in siblings' outcomes. Alternatively, parents may choose to direct more

resources to children with higher endowment at birth ('reinforcing behavior'). Consequently, this can amplify the differences between offspring born with different birth endowments. To this end, using administrative data from Chile, the authors find empirical evidence in support of a compensatory framework. Further, using the National Longitudinal Survey data, Restrepo (2016) adds to the existing empirical evidence on parental response to children's birth endowment. The author finds that parents' compensating behavior can vary by their socio-economic characteristics such as mothers' education. These empirical findings hold relevance to our research objective. This is because, with respect to children's bodyweight during preschool years, the evidence of 'catching-up' by smokers' children (who are born with lower birthweight) with children of non-smoker mothers can potentially be attributed to parents' compensating behavior which can further be explained by unobserved attributes such as innate characteristics or attitudes. If this argument holds true, the literature evidence linking maternal smoking during pregnancy with child risk of being overweight can be argued to be associational rather than causal.

In general, it appears clear that compared to the strong causal evidence with regard to children's birthweight, the empirical findings in terms of impact of maternal smoking during pregnancy on body weight during childhood years are mostly descriptive in nature. We therefore contribute to the current knowledge base by combining the two aforementioned strands of the literature and exploring the validity of the catch-up phenomenon using a panel of children born to a nationally representative sample of American mothers. We employ established empirical techniques to exploit exogenous variation in maternal smoking behavior to test if there is a causal influence of prenatal exposure to maternal smoking on the risk of being overweight during pre-school ages.

3. DATA: THE NATIONAL LONGITUDINAL SURVEYS

We link mothers' information from the original cohort of the NLSY79 with information on their biological children from the NLS-CYA. Administered by the Bureau of Labor Statistics, the National Longitudinal Surveys are widely utilized for social and policy-relevant empirical research. The surveys incorporate a wide range of socio-economic and demographic information on a nationally representative sample of 12,686 individuals, who were born between 1957 and 1964. Commencing in 1979, the surveys were administered annually until 1994, and biennially thereafter. The NLS-CYA commenced in 1986 and are conducted biennially. These surveys document health, schooling, behavioral, as well as family-specific information of biological children born to women in the original NLSY79 cohort.

Upon linking the two surveys and conducting a list wise deletion based on all relevant variables, our total matched sample includes 7,174 children born to 3,881 mothers. However, depending on the availability of the data required to test the relationship of interest, we apply further restrictions to this sample.

Our main analysis has two elements: (i) impact of mothers' smoking during pregnancy on children's birthweight, and (ii) impact of mothers' smoking during pregnancy on bodyweight of pre-school children aged two to five.³ For the first part of our analysis, we limit our sample to children who were born between 1979 and 2011 (the last year in which a childbirth was recorded in the NLS-CYA). This is done in order to maximize our sample size and simultaneously control for important mother- and family-level covariates derived from the NLSY79 (which commenced in the year 1979). Further, since the NLSY79 were conducted biennially since the 1994 survey, for children born during odd-numbered years post-1994 i.e.

-

³ By the standard approach in child health literature, especially with respect to US-based studies, early childhood obesity is usually evaluated from the age of two (Ogden et al. 2008; 2010; 2012). To maintain consistency with approach in the related literature, we exclude children aged one from our main analysis.

from 1995 onwards until 2011, the covariates derived from the NLSY79 are based on information from survey years immediately succeeding the birth years. At this point, it is important to note that excluding children with odd-numbered birth year post-1994 survey does not affect our key findings in the birthweight analysis. Moreover, these children account for marginally less than 5 percent of the birthweight regression sample.

For the second part of our empirical exercise, investigating the impact on pre-school age bodyweight, we begin by including the same set of children included in the birthweight analysis. However, not all children included in the birthweight analysis were surveyed at every age from two through five, due to the biennial nature of NLS-CYA. Therefore, depending on a child's birth year, their later childhood information are either documented at odd numbered or even numbered ages. As a result, for post-birth years, we have one set of children who have been surveyed at the ages of two and four, and another set of children surveyed at ages three and five. As indicated earlier, we do not include children aged one. The study period for the preschool analysis spans across the biennial survey years between 1986 and 2012.

For our preferred specification (i.e. 2-SLS), in conducting the preschool weight analysis, we pool all children aged two to five together and treat each survey unit as a distinct observation such that the data resembles a cross-sectional sample. However, we also take advantage of the panel aspect of the data to verify whether our key findings our supported by alternative empirical specifications that account for mothers' fixed effects by exploiting 'within-mother across child birth' variation in smoking behavior (Rosenzweig & Wolpin, 1991).

The main outcome variable in our analysis is children's bodyweight reported in NLS-CYA's pre- and post-natal information (expressed in kilograms). The key explanatory variable is a binary indicator that incorporates mother's smoking behavior during pregnancy. In our most parsimonious model for the birthweight analysis, we additionally control for a wide range of

mother and child demographic, social and health-specific contemporaneous characteristics. The mother and family-specific information are captured by indicator of mothers' prenatal substance use (alcohol or other illicit drugs including marijuana, cocaine, etc.), mother's age, height, marital status, employment status, and family's poverty status at childbirth. The child-specific characteristics include child's sex, race, birth order, and birth length. We later test the robustness of our 2-SLS estimates in a more saturated version of the above specification by incorporating additional controls in our regressions.

Similar controls are incorporated in the preschool age bodyweight analysis. More importantly with respect to preschoolers' analysis, since our objective is to test the effect of in-utero exposure of maternal smoking on children's likelihood of being overweight during the early childhood years, we create two dichotomous indicators of being overweight using World Health Organization's (WHO) body mass index-for-age (BMI-for-age) and weight-for-age distributions. In addition to continuous measure of children's weight during preschool years (ages 2-5), we consider these two binary overweight indicators as our dependent variables of interest in the preschool analysis.

To construct the overweight indicators using standard definitions, we assign a value one when child's BMI and weight lie at or above 85th percentile in relevant distributions and zero otherwise. One of the empirical advantages of using these binary bodyweight indicators is that the construction of the variables is performed using age- and sex-specific distributions. This mitigates comparability concerns with respect to children's health outcomes due to differences across children's sex and age, and therefore allows us to maximize our analysis sample in the preschool analysis. Pursuant to our earlier discussion (in Section 2), the majority of studies that evaluate the association between prenatal smoking and the risk of child being overweight use

similar bodyweight measures (additional example includes Wang & Lim, 2012).⁴ Using WHO's BMI and weight-for-age distributions, we further create two binary indicators of being underweight that are assigned a value of one when child's BMI or weight lies below the 5th percentile of the two corresponding distributions.

Additionally, our access to restricted NLSY data permits control for state-level time-invariant characteristics through state fixed effects in both sets of analyses. Tables 2 and 3 provide descriptive information on all relevant variables used in our analysis. For ease of comparison, we present the summary statistics for samples separated by mothers' smoking behavior during pregnancy. We discuss the main takeaways from these tables later in Section 5.

4. IDENTIFICATION STRATEGY

4.1 Cross-sectional analysis

Our cross-sectional regression analyses employ 2-SLS regressions to evaluate causal effects of maternal smoking during pregnancy on child health outcomes of our interest. With respect to the birthweight analysis, our IV regression model is-

$$Y_{is} = \alpha_1 + \delta. Mother's Smoking_{is} + \alpha_2. X_{is} + \epsilon_{is}$$
 (1)

$$Mother's Smoking_{is} = \beta_1 + \gamma \cdot tax_s + \beta_2 \cdot X_{is} + v_{is}$$
 (2)

where Y_{is} represents weight of child i from state s at birth. Mother Smoking is is an indicator of prenatal smoking participation of mother of child i. X_{is} is a vector of child- and mother-specific characteristics including state fixed effects.

To study the association between maternal smoking and children's birth weight, we first

-

 $^{^4}$ The formula utilized in calculating BMI (in kg/m²): BMI = Weight / (Height in inches)²; See https://www.bcbst.com/providers/MPMTools/BMICalculator.shtm.; Accessed on July 17, 2018.

estimate ordinary least squares (OLS) regression represented by equation (1). Next, to account for potential biases in the OLS estimates resulting from exclusion of unobserved heterogeneities, we estimate 2-SLS regressions whose first-stage is represented by equation (2).

In our analysis, to be treated as an instrument, variable tax_s should be correlated with mothers' smoking behavior (during pregnancy) but uncorrelated with unobserved heterogeneities that can influence children's health outcomes. Based on previous studies, we expect that a variable that captures exogenous variation in state and federal tobacco tax rates over the years meets these requirements (Evans & Ringel, 1999; 2001; Lien & Evans, 2005; Simon, 2016). To construct our IV, we consider the sum of state and federal tobacco tax rates during the year a mother likely conceived her child. The conception year was calculated by using information on children's reported gestation length (originally reported in weeks) and birth date (in month and year) in NLS-CYA. The state-level variation in the sum of the two tax rates across pregnancy years allows us to estimate the effect of the governmental tobacco regulations on mothers' smoking participation during pregnancy (Evans & Ringel, 1999).

For the later childhood bodyweight analysis (ages 2-5 years), we adopt a similar empirical approach that resembles the birthweight specifications given by equations (1) and (2).

4.2 Within mother across childbirth regressions

An improvement over the OLS model discussed in 4.1 would be to account for mother-specific time-invariant unobserved factors and estimate a mother fixed effects regression as below:

$$Y_{cm} = \rho_1 + \lambda. Mother Smoking_{cm} + \rho_2. X_{cm} + \mu_m + u_{cm}$$
 (3)

where Y_{cm} represents birthweight (in kilogram) of child c born to mother m.

Mother $Smoking_{cm}$ is the binary indicator of whether a mother m smoked during pregnancy

before birth of child c. X_{cm} is a vector of child- and family-specific characteristics that vary within each cluster of mothers including variables that vary either by birth year or just across sibling (such as child's birth order). μ_m represents mother-specific time-invariant effects that are likely to be correlated with their smoking behavior and may affect child health outcomes. It is further important to note that μ_m can also be treated as family or sibling fixed effects given the nature of the survey design. Because we control for mothers' FE, our identification strategy relies on within mother and across pregnancies variation in smoking behavior (Rosenzweig & Wolpin 1991). Finally, u_{cm} is the error term. In case the source of bias in the OLS estimate is generated only from unobserved time-invariant mother and family-specific heterogeneities, the estimated parameter λ in equation (3) would represent an unbiased measure of the impact of maternal smoking on children's birthweight. However, the FE estimates may be biased if there are time-variant unobserved characteristics that are correlated with both smoking behavior and children's health outcomes. While, this particular empirical concern is likely to be addressed by instrumental variable strategy described in the previous subsection, estimation of mother FE regressions provides additional evidence to check the robustness of our 2-SLS regression findings. However, it is important to note that the treatment effects evaluated by the 2-SLS models and mother FE regressions differ in the way that the former looks at between differences in mothers' smoking behavior during pregnancy and the latter estimates the withinmother variation in prenatal smoking. Further, since identification of FE estimate relies on within mother across pregnancies variation in smoking behavior, we restrict our FE analysis to mothers with two or more children only.

5. RESULTS

5.1 Summary statistics of variables

We begin by discussing the descriptive statistics of the two main samples we use in our analysis. Table 2 presents summary statistics for the child sample at birth. We also present

statistics for the overall child sample, and its subsamples separated by gender and smoking participation of mothers during pregnancy. Average birth weight of children is found to be 3.32 kg, which lies in the range of birthweight estimates reported in other studies (see Table III in Butler et al., 1972). In the overall child sample, we find that 28 percent of mothers reported to have smoked during pregnancy and further 32.5 percent reported to have used some substance during pregnancy including alcohol, marijuana, cocaine or other illicit drugs. Demographically, approximately 57 percent of women identify as White, 25 percent as Black, and 18 percent are Hispanic. Female children make up about 49 percent of the full sample.

With respect to socio-economic and health-related conditions, we present descriptive information of variables including mothers' employment status, marital status, highest academic qualification, and family's poverty status at the time of childbirth. Based on these variables, we find that 73 percent of mothers report being employed at the time of birth and 68 percent mothers are married at the time birth (this excludes women who were previously married but do not currently have a partner). Further, while 44 percent have a high school degree and 34 percent reported to have went to college. Focusing of family's economic condition, we see 23 percent of our sample fall below the Federal poverty threshold. These statistics are similar across samples of boys and girls (reported in columns 4 and 7 respectively).

To provide further understanding of socio-economic and demographic differences between smokers and non-smokers, we present descriptive information of all relevant variables in Table 2 classified by mothers' smoking behavior during pregnancy. Consistent with our expectation based on previous literature we find that children's average birth weight in smoking sample is less than that in nonsmoking sample (columns 2 and 3). In particular, the difference in children's average birthweight in the two samples is equivalent to 240 grams. This finding

further holds in the separate samples of boys (columns 5 and 6) and girls (columns 8 and 9).

Demographically, while the share of Whites appears to be substantially larger in the smokers' sample (66 percent of smokers versus 53 percent of non-smokers), the proportion of Hispanics appear to be comparatively higher in the non-smokers' sample (10 percent of smokers versus 22 percent of non-smokers). In addition, mothers who smoke during pregnancy are less likely to be employed or have gone to college, and more likely to be under federally defined poverty line. Mothers who smoked during pregnancy are also found to be more likely to have engaged in prenatal consumption of alcohol or other illicit substance.

Regarding our instrumental variable, the average federal and state-level tax rates during pregnancy years are US\$ 0.32 and 0.25 per pack of 20 cigarettes respectively. These tax rates are converted in 2005 dollars to account for the effects of price inflation.⁵

⁵ The tax rate data have been accessed from 'Tax Burden on Tobacco' (Orzechowski & Walker, 2017). Also see https://healthdata.gov/dataset/tax-burden-tobacco-1970-2018 for state and federal tax rates in terms of US\$ per pack of 20 cigarettes; Accessed on April 30, 2018.

Table 2

Mean/proportions of variables used in birthweight regressions

		All childre	n		Boys only			Girls onl	y
	Total	Smoker	Non-	Total	Smoker	Non-	Total	Smoker	Non-
			smoker			smoker			smoker
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
								• • •	
Birth weight (in kg)	3.32	3.15	3.39	3.38	3.22	3.44	3.26	3.09	3.33
Prenatal smoking (fraction)	0.28	-	-	0.28	-	-	0.28	-	-
Prenatal substance use	0.33	0.48	0.27	0.33	0.49	0.27	0.32	0.47	0.26
Birth length (in inches)	20.02	19.76	20.12	20.18	19.86	20.30	19.86	19.66	19.93
Birth order	1.99	2.11	1.95	1.98	2.09	1.94	2.00	2.13	1.95
White	0.57	0.66	0.53	0.57	0.66	0.53	0.57	0.66	0.53
African-American	0.25	0.24	0.25	0.24	0.23	0.25	0.25	0.25	0.26
Hispanic	0.19	0.10	0.22	0.19	0.11	0.22	0.18	0.09	0.22
Female child	0.49	0.49	0.49	-	-	-	-	-	-
Mother's age	26.40	25.35	26.81	26.40	25.29	26.83	26.40	25.42	26.78
Employed	0.73	0.69	0.75	0.73	0.67	0.74	0.75	0.71	0.76
Married	0.68	0.54	0.73	0.69	0.54	0.74	0.67	0.54	0.73
High-school	0.44	0.49	0.42	0.44	0.47	0.42	0.44	0.50	0.42
College	0.34	0.16	0.41	0.34	0.16	0.41	0.34	0.15	0.41
Poverty status	0.23	0.33	0.20	0.23	0.33	0.19	0.24	0.33	0.20
Mother's height (in meters)	1.63	1.63	1.63	1.63	1.63	1.63	1.63	1.64	1.63
Tobacco Tax	0.58	0.56	0.58	0.58	0.56	0.58	0.58	0.56	0.58
Federal cigarette tax	0.25	0.24	0.25	0.25	0.24	0.25	0.25	0.24	0.25
State cigarette tax	0.32	0.31	0.33	0.32	0.31	0.33	0.32	0.31	0.33
Observations	7174	2009	5165	3670	1030	2640	3504	979	2525

Notes: The above table presents the sample proportions/ averages of all variables used in our birthweight regression analysis. The Federal and state-level cigarette tax rates are in terms of 2005 dollars/ per pack of 20 cigarettes (Tax burden on tobacco (Orzechowski & Walker, 2017)).

Table 3 presents summary statistics for our preschoolers' sample (related to children aged 2 to 5). Focusing on child characteristics, in the overall sample of preschoolers (see column 1), the average body weight is approximately 15.6 kg and average height is 0.98 meters. Further, based on weight-for-age measures, we find that 19 percent of children lie above 85th percentile of their relevant distribution. However, with the respect to BMI-for-age, 30 percent of children appear to be above 85th percentile of their corresponding distribution. Of particular relevance to our study is the finding that the proportion of children in the smokers' group who are found to be in the BMI-based overweight category (32 percent) is marginally higher than the relevant proportion in the non-smokers' group (29 percent). This aligns with the literature evidence on the positive association between prenatal smoking and likelihood of children being overweight.

In addition, the average proportions of children having bodyweight measures lying below 5th percentile of weight-for-age and BMI-for-age distributions are 6 and 8 percent respectively. Interestingly, in contrast to the birthweight analysis sample, we do not observe any noteworthy differences in the mean proportions of being overweight when the preschoolers' overall sample is classified by children's sex (columns 4 and 7) and mothers' prenatal smoking behavior (columns 2-3; 5-6; and 8-9).

The preschoolers' sample estimates of socio-economic and demographic characteristics of mothers do not differ much from those observed in the childbirth sample. Among the mothers in the preschoolers' sample 53 percent identify themselves as White, 26 percent are African-Americans, and 20 percent are Hispanic. Additionally, about 70 percent of mothers report being employed when at the time of survey, and approximately 69 percent are married. We do not see any major differences in the sample means of the mothers and family level variables in boys and girls only subsamples (see columns 4 and 5 respectively).

Table 3

Mean/proportions of variables used in preschool weight regressions (2-5 years)

	All children				Boys only		Girls only			
	Total	Smoker	Non-	Total	Smoker	Non-	Total	Smoker	Non-	
			smoker			smoker			smoker	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	
Ol'11 ' 1, (' 1)	15.50	15.67	15.56	15.04	15.00	15.00	15.22	15 42	15.20	
Child weight (in kg)	15.59	15.67	15.56	15.84	15.90	15.82	15.33	15.43	15.29	
Overweight (BMI)	0.30	0.32	0.29	0.33	0.35	0.32	0.27	0.28	0.26	
Underweight (BMI)	0.08	0.08	0.09	0.08	0.07	0.08	0.09	0.09	0.09	
Overweight (WFA)	0.19	0.19	0.19	0.20	0.19	0.20	0.17	0.18	0.17	
Underweight (WFA)	0.06	0.07	0.06	0.06	0.07	0.05	0.06	0.07	0.06	
Prenatal smoking (fraction)	0.28	-	-	0.28	-	-	0.28	-	-	
Prenatal substance use	0.34	0.49	0.29	0.36	0.49	0.30	0.33	0.48	0.27	
Child height (in meter)	0.98	0.97	0.98	0.98	0.97	0.98	0.97	0.98	0.97	
Birth order	2.02	2.10	1.99	2.02	2.11	1.99	2.01	2.09	1.98	
Child age	3.52	3.58	3.49	3.52	3.57	3.50	3.52	3.60	3.49	
White	0.53	0.65	0.49	0.53	0.64	0.49	0.54	0.66	0.49	
African-American	0.26	0.25	0.27	0.26	0.25	0.26	0.27	0.26	0.28	
Hispanic	0.21	0.10	0.25	0.22	0.11	0.26	0.19	0.09	0.24	
Female child	0.49	0.49	0.49	-	-	-	-	-	-	
Mother's age	28.69	28.06	28.94	28.75	28.12	29.00	28.63	28.00	28.88	
Employed	0.68	0.65	0.69	0.67	0.63	0.69	0.69	0.67	0.70	
Married	0.69	0.57	0.73	0.69	0.57	0.74	0.68	0.57	0.72	
High-school	0.49	0.52	0.48	0.50	0.51	0.49	0.49	0.54	0.47	
College	0.31	0.17	0.37	0.31	0.17	0.36	0.32	0.18	0.37	
Poverty status	0.28	0.38	0.24	0.28	0.39	0.24	0.28	0.37	0.24	
Household size	4.39	4.28	4.43	4.39	4.29	4.43	4.38	4.27	4.42	
Mother's height (in meters)	1.63	1.63	1.63	1.63	1.63	1.63	1.63	1.64	1.63	
Tobacco Tax	0.56	0.54	0.56	0.56	0.55	0.56	0.56	0.54	0.57	
Federal cigarette tax	0.25	0.25	0.26	0.25	0.25	0.26	0.25	0.25	0.26	
State cigarette tax	0.30	0.30	0.30	0.30	0.30	0.30	0.30	0.30	0.31	
Observations	4991	1391	3600	2548	713	1835	2443	678	1765	

Notes: The above table presents the sample proportions/ averages of all variables used in our pre-schoolers' regression analysis. The Federal and cigarette tax rates are in terms of 2005 dollars/ per 20 cigarettes (Tax burden on tobacco (Orzechowski & Walker, 2017)). The age and sex-specific binary indicators are constructed using World Health Organization's standard thresholds of body mass index and weight-for-age distributions of boys and girls. The information has been accessed from https://www.who.int/childgrowth/standards/en/. Accessed on April 23, 2017.

5.2 Birthweight analysis

Table 4 presents estimates of effects of maternal smoking during pregnancy on children's birthweight, as estimated by the baseline OLS as well as 2-SLS and mother FE regression models.

With respect to OLS regression estimates, for the overall sample that controls for child's sex in addition to other relevant covariates (column 1), we find that on average, birthweight of children whose mothers smoked during pregnancy falls short of birthweight of children whose mothers did not by 0.18 kilograms (kg). We also see that the OLS regression estimate is statistically significant at the one percent level. We find comparable reductions in birth weight when comparisons are made across same child gender. Specifically, we observe that mothers' smoking during pregnancy is associated with a reduction of 0.15 kg in birthweight when comparisons are made among boys only (column 4) and 0.19 kg in the girls' sample (column 5).

As indicated earlier, the main empirical concern regarding OLS regressions is failure to account for all observed and unobserved individual, family-level and socio-economic characteristics that may be simultaneously associated with mothers' smoking and child's well-being. To address these endogeneity concerns, we use an IV strategy illustrated in Section 4. For empirical validity of our instrument, the cigarette tax-based instrument should meet two basic requirements. First, the instrument must be strongly correlated with mother's smoking behavior (the relevance criteria), and second, the instrument must be uncorrelated with unobserved heterogeneities that affect the child's body weight except through the channel of mother's smoking, conditional on all controls included in model (the excludability requirement).

We argue that the federal and state taxes levied on cigarettes over years meet the above two requirements. However, while the empirical evidence in support of the first assumption is presented in Table 4 itself, where we present our first stage estimate of 2-SLS regressions, the exclusion restriction criteria cannot be directly tested.

One of the concerns regarding the validity of the excludability assumption is that state-level tobacco tax policy may be correlated with implementation of other health-promoting welfare programs, which may have impact on children's weight. For example, states with high tobacco tax rates may also be more likely to have high soda tax (e.g. New York). Because such possibilities cannot be ignored, in all our regression models, we control for state fixed effects (mothers' state of residence at the time of survey) that account for any time-invariant differences across states. In addition, ex ante, another possibility is that tobacco taxes might be large enough to influence other health-related behaviors that we do not control for in our main regression models. Therefore, in a supplemental robustness exercise (presented in Appendix Table A.1), we estimate our IV model controlling for additional prenatal controls such as prenatal visits and vitamin intake.

With respect to the first stage results of our 2-SLS regressions with children's birthweight as dependent variable, we find that pregnant women respond strongly to tax increases. In particular, a dollar increase in real tobacco tax rate results in a decline in mothers' probability of smoking during pregnancy by 26 percentage points (column 3). These results are qualitatively consistent with findings in prior studies that evaluates the effect of tobacco taxes on maternal smoking during pregnancy (Evans & Ringel, 1999; 2001; Lien & Evans, 2005). Further, in the gender-specific samples, the decline in the probability of smoking during pregnancy as a result of state and federal tobacco taxes is found to be larger in case of boys sample relative to that of girls (0.29 versus 0.21 percentage points; columns 5 and 8). In all three samples, we find that the instrumental variables are strong predictors of maternal smoking during pregnancy, as indicated by the statistically significant (at the one percent level) first-

stage regression coefficient values. Moreover, for the overall child sample and for the sample of boys, the partial F-statistics are 30.04 and 21.34, which are above the recommended value of 10 (Stock, Wright, and Yogo, 2002).

Focusing on the second-stage results, we find that prenatal smoking leads to a decline in children's birthweight by 0.53 kg (531 grams; column 2). The estimated coefficient is statistically significant at the 5 percent level. While the 2-SLS estimate is larger than our OLS estimate, it is important to note that IV estimates represent Local Average Treatment Effect (LATE) rather than Average Treatment Effect (ATE). More specifically, in our case, the 2-SLS regression generates estimates of the effect of maternal smoking on children born to mothers who are at the margin of adjusting their smoking behavior based on variation in tobacco taxes. Nonetheless, our 2-SLS estimate of the effect of maternal smoking during pregnancy on children's birthweight lies within the range of IV-based point estimates obtained by Evans & Ringel (1999) who utilize the Natality Data Files (1989-1992) for their analysis. In particular, Evans & Ringel's (1999) 2-SLS estimate of the effect of mothers' smoking participation birthweight varies from 0.36 to 0.59 kg depending on a step-by-step inclusion of additional covariates ranging from a less saturated specification to a more highly saturated model. In the gender-specific sample, we find comparing birthweight among boys only, maternal smoking during pregnancy leads to 0.63 kg decline in birthweight. However, the regression coefficient is statistically significant at the 10 percent level. On the other hand, in the girls only sample, the second stage regression coefficient is not statistically indistinguishable from zero although the coefficient is negative.

In our final empirical specification, we control for mother fixed effects by estimating equation (3). The FE estimate of birth effect as seen in column 3 of Table 4 suggests that mothers' prenatal smoking leads to a reduction in children's birthweight by 0.15 kilogram. Once again,

our FE estimate of the impact of maternal smoking during pregnancy on children's birthweight lies in the range of the FE estimates presented in Rosenzweig & Wolpin's (1991) study. To illustrate, Rosenzweig & Wolpin's (1991) FE estimate represented a decline in children's birthweight by 2.92 ounce (≈ 0.08 kg) for prenatal smoking amounting to less than a pack and 5.62 ounce (≈ 0.16 kg; for prenatal smoking amounting to a pack or more). With respect to our FE regressions using gender-specific samples, we find qualitatively similar results as that of our 2-SLS analysis. In other words, we find statistically significant and relatively larger effects for boys (represented by a decline in birthweight by 0.26 kg) compared to that of girls (see columns 6 and 9).

Table 4

Estimates of the effect of maternal smoking during pregnancy on birthweight

		All			Boys only			Girls only		
	OLS	2-SLS	FE	OLS	2-SLS	FE	OLS	2-SLS	FE	
Sample mean	3.	32	3.33	3.	.38 3.38		3.	3.27		
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	
Prenatal smoking	-0.176*** (0.016)	-0.531** (0.247)	-0.147*** (0.035)	-0.157*** (0.023)	-0.632* (0.301)	-0.263*** (0.068)	-0.190*** (0.021)	-0.281 (0.428)	-0.052 (0.055)	
First stage (Depend Sample proportion	dent variabl	e: Indicator 0.28	of smoking	g during pre	egnancy) 0.28			0.28		
Tobacco tax		-0.258*** (0.047)			-0.294*** (0.062)			-0.207*** (0.067)		
Partial F	-	30.03	-	-	21.34	-	-	9.36	-	
Observations	7174	7174	5543	3670	3670	2842	3504	3504	2701	

Notes: The above table presents regression estimates obtained from ordinary least squares (OLS), two-stage least squares (including the first stage estimates), mother fixed effects (FE) regressions for all children (1-3), boys only (4-6), and girls only (7-9) samples. All models control for child characteristics, mother characteristics, and state fixed effects. Robust standard errors are corrected for clustering on the mothers and are presented in parenthesis. ***, **, * denote the coefficients are significantly different from zero at the 10%, 5%, and 1% level respectively.

5.3 Bodyweight outcomes of preschoolers

For children aged between 2 and 5, we begin by analyzing children's weight (expressed in kg) as our dependent variable. It is however important to note, that continuous measure of children's weight may lack comparability owing to potential physiological differences between genders and across ages. Nonetheless, prior to focusing on the binary health measures of being overweight, analyzing children's weight shall allow us to examine whether on average, preschoolers' weight continues differs during early childhood years, as observed in the birthweight analysis. Table 5 presents our regression estimates obtained from all three empirical specifications - OLS, 2-SLS and mother FE models.

In all our child samples (columns 1, 4, and 7), the OLS estimates suggest that there is no statistically significant association between maternal smoking during pregnancy and children's weight during preschool years. However, in the 2-SLS regressions, we observe that weight of children of non-smokers continue to exceed weight of smokers' children. For instance, in the overall sample, the second-stage regression estimate indicates a gap of 4.40 kg between the two groups of interest. This effect is statistically significant at the 1 percent level. Further for boys, the negative effect of maternal smoking on children's weight (during preschool years) remains larger than that of girls, as indicated by a difference of 5.79 kg (for boys) versus a difference of 2.97 kg (for girls). Finally, the first stage coefficients are largely similar to the first-stage results obtained in the birthweight IV analysis.

In relation with mother fixed effects regressions for preschoolers, it is first important to discuss the empirical adjustments we applied to the preschoolers' sample used in the OLS and 2-SLS analyses. By construction, OLS and 2-SLS estimation allow the preschooler's sample to be treated as cross-sectional data. To put it more succinctly, despite the possibility that a child-mother pair can be observed repeatedly given the longitudinal nature of the biennial NLS-CYA

data, information recorded for a particular child-mother pair at each survey year can be treated as a unique observation. However, in case of mother fixed effects regression where we try to exploit variation in mothers' prenatal smoking (across multiple childbirths), observing a child-mother pair multiple times in a longitudinal data format may lead to misspecification due to lack of variation in our key time-invariant regressor i.e. mothers' smoking participation during pregnancy. Therefore, to avoid double counting of a child-mother pair, we compress our OLS (or 2-SLS) sample by each child so that the data format resembles the format used in the birthweight analysis where we have a unique observation for each mother-child pair. The data compression is performed in a way such that in the resultant sample, all the time-variant variables used in the mother fixed effects regressions can be interpreted as an average value of the respective indicators over the children's preschool years (2-5 years) under evaluation.

Based on Table 5 estimates, our mother's fixed effects regression using the resultant overall child sample indicates that maternal smoking does not have any significant impact on preschoolers' weight (see column 3).

Table 5

Estimates of the effect of maternal smoking during pregnancy on pre-schoolers' weight

		All			Boys only			Girls only		
	OLS	2-SLS	FE	OLS	2-SLS	FE	OLS	2-SLS	FE	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	
Sample mean	15	5.52	15.50	1:	5.82	15.78	15	.22	15.21	
Prenatal smoking	0.063 (0.101)	-4.400*** (1.588)	0.341 (0.225)	0.078 (0.152)	-5.792** (2.927)	0.871* (0.474)	0.060 (0.132)	-2.969* (1.722)	-0.054 (0.415)	
First stage (Depend Sample proportion	lent variab	ole: Indicato 0.28	r of smok	zing durin	g pregnanc 0.28	ry)		0.28		
Tobacco tax		-0.300*** (0.055)			-0.265** (0.088)			-0.329*** (0.067)		
Partial F	-	29.26	-	-	8.88	-	-	24.01	-	
Observations	4553	4553	2533	2548	2548	1274	2239	2239	1259	

Notes: The above table presents regression estimates obtained from ordinary least squares (OLS), two-stage least squares (including the first stage estimates), mother fixed effects (FE) regressions for all children (1-3), boys only (4-6), and girls only (7-9) samples. All models control for child characteristics, mother characteristics, and state fixed effects. Robust standard errors are corrected for clustering on the mothers and are presented in parenthesis. For the FE regression samples, we collapse all the variables at their respective means such that each observation relates to a unique mother-child pair to avoid lack of variation in maternal smoking during pregnancy. ***, **, * denote the coefficients are significantly different from zero at the 10%, 5%, and 1% level respectively.

Next, we focus on the binary indicators of being overweight. While bodyweight status of being overweight or obese is usually evaluated using BMI measures, we use a second overweight indicator (based on weight-for-age distribution) for additional evidence. Further, since the dichotomous indicators are evaluated by accounting for child age- and sex-specific distributions of respective weight measures, we perform our analysis only on the overall child sample used in Table 5 regressions. However, we do control for children' sex and age in our regressions in addition to other relevant covariates.

In Table 6, consistent with the evidence in prior health literature discussed earlier (see Section 2), in our linear probability analysis of BMI-based overweight indicator (column 1), we find that maternal smoking during pregnancy is associated with a 3 percentage points increase in the probability of being overweight. However, considering exogenously determined smoking participation during pregnancy, our 2-SLS estimates indicate prenatal smoking does not have any statistically significant influence on the risk of children being overweight (column 2). This result holds when we consider our second overweight indicator (column 5). However, based on our IV analysis, we also do not find any statistically significant effect of prenatal smoking on the likelihood of being underweight among preschoolers (columns 8 and 11). ⁶

Finally, using the compressed child sample used in the FE analysis reported in Table 5, we find that once we account for mothers' unobserved time-invariant fixed effects, maternal smoking during pregnancy is negatively associated with both BMI-based indicators of being overweight and underweight, and weight-for-age indicator of being underweight. It should be noted that in

-

⁶ To further test the 'catch-up' growth phenomenon discussed in the prior literature, we estimate OLS and mother FE specifications using children's weight as dependent variable on child samples for each preschool age (from 2 to 5 years). Although we do not intend to overstate the importance of the age-specific findings due to small sample size at each age, we present these results in Appendix Table A.2. Our FE estimates indicate no statistically significant differences in children's weight between the smoking and non-smoking group at any of the preschool ages. We further refrain from estimating 2-SLS regressions as the second-stage regression coefficients are imprecisely estimated owing to the lack of the IV's predictive power in the small age-specific child samples.

the compressed sample, the dependent variable could be interpreted as proportion of times a child is observed in the survey to be overweight or underweight (bounded by a 0-1 interval) during the preschool ages under evaluation. Owing to the fractional form of our dependent variable, the reliability of linear FE estimates may be restricted due to possible non-linearity in the effects of our regressor (Papke & Wooldridge, 2008). Further, lack of continued availability of relevant information for all children in NLS-CYA in consecutive surveys may also misrepresent the actual proportion of times a child is observed to be in a particular bodyweight category.

Overall, our analysis using the national longitudinal surveys indicates that once confounding influences of unobserved heterogeneities are accounted for, maternal smoking does not have any causal impact on the risk of being overweight among preschool aged children. This is contrary to the discussions presented in the prior literature based on descriptive empirical evidence that guide us to conjecture that maternal smoking during pregnancy leads to childhood obesity. As such our findings introduce a wide scope for future research to explore potential mechanisms and socio-economic factors or individual characteristics (including lifestyle choices and decisions made by parents) that may underlie the observed positive association between maternal smoking during pregnancy and risk of having excess weight during early childhood (Simon 2016).

⁷ We do test our FE estimates using generalized linear models, following Papke & Woolridge's (2008) recommendation on panel data treatment of fractional response variables (also see Wooldridge's (2011) Stata conference discussion). Findings are qualitatively similar to the FE results. Since, this additional empirical test is not directly pertinent to our analysis results from generalized linear model we do not present our findings but results are available upon request.

Table 6

Estimates of the effect of maternal smoking during pregnancy on pre-schoolers' bodyweight indicators

	Overweight (BMI)			Overweight (Weight-for-age)			J	Inderweig (BMI)	ht		ght age)	
	LPM	2-SLS	FE	LPM	2-SLS	FE	LPM	2-SLS	FE	LPM	2-SLS	FE
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)
Sample proportion	0.	.30	0.28	0.1	18	0.17	0.	08	0.08	0.	06	0.05
Prenatal smoking	0.029* (0.017)	-0.157 (0.287)	-0.063** (0.027)	0.004 (0.014)	-0.056 (0.171)	0.033 0.041	-0.008 (0.010)	-0.055 (0.134)	-0.062** (0.027)	0.010 (0.008)	-0.209 (0.164)	-0.037* (0.020)
Partial F stat	-	14.51	-	-	29.26	-	-	41.60	-	-	29.26	-
Observations	4553	4553	2533	4553	4553	2533	4553	4553	2533	4553	4553	2533

Notes: The above table presents regression estimates obtained from linear probability model (LPM), two-stage least squares (including the first stage estimates), mother fixed effects (FE) regressions for indicators of overweight and underweight based on BMI and weight-for-age distributions. All models control for child characteristics, mother characteristics, and state fixed effects. Robust standard errors are corrected for clustering on the mothers and are presented in parenthesis. For the FE regression samples, we collapse all the variables at their respective means such that each observation relates to a unique mother-child pair to avoid lack of variation in maternal smoking during pregnancy. Consequently, the dependent variable for FE regressions represent the proportion of times a child is observed to be overweight or underweight. ***, **, * denote the coefficients are significantly different from zero at the 10%, 5%, and 1% level respectively.

6. CONCLUSION

The numerous health hazards associated with smoking demonstrate its severity on the public health front. In fact, cigarette smoking is known to harm almost every organ of the human body, cause several diseases, and reduce the health of smokers in general, making it the leading preventable cause of death in the United States (Centers for Disease Control and Prevention, 2013). Moreover, the health effects of cigarette smoking reach far beyond the smokers themselves. This study provides new and policy-relevant evidence on the child health impacts of maternal smoking during pregnancy. Our contribution to the literature in this space is a focus on the causal link between prenatal exposure to maternal smoking and future child weight outcomes.

In summary, while maternal smoking is found to have a significant negative impact on children's birth weight, the same does not cause a higher risk of obesity during preschool ages. Our analysis indicates that the associational evidence on the positive link between maternal smoking and child obesity may be mediated by poor socio-economic, health, and behavioral characteristics that can commonly be attributed to mothers who smoke during pregnancy. Assuming that these characteristics are driven by mothers' unobserved characteristics, our instrumental variable strategy corrects for the potential confounders and produces plausibly more reliable estimates.

This study provides enhanced motivation for implementing policies to effectively address smoking behavior among pregnant mothers. In addition to continuing with traditional ways to curb smoking such as through information campaigns, smoke-free policies, age restrictions, etc., there is a need to implement broader and more effective methods to motivate quitting. For example, a recent paper by Islam, Folland & Kaarbøe (2017) provides evidence on how investing in social capital variables such as community trust and participation in organization

activities could lower the incidence of smoking.⁸ Finally, our results underscore the role of socioeconomic causes of childhood obesity and indicate the need for effective intervention in this area of public health.

⁸ There has been some evidence that quitting smoking may be associated with increased risk of obesity (see Chou et al, 2004 and Liu et al, 2010). However, recent evidence by Pieroni & Salmasi (2016) maintain the importance of policies aimed at reducing smoking, as they find limited consequences in terms of a rise in obesity risk among adults.

REFERENCES

- 1. Al Mamun, Abdullah, Debbie A. Lawlor, Rosa Alati, Michael J. O'callaghan, Gail M. Williams, and Jake M. Najman. "Does maternal smoking during pregnancy have a direct effect on future offspring obesity? Evidence from a prospective birth cohort study." *American Journal of Epidemiology* 164, no. 4 (2006): 317-325.
- 2. Bharadwaj, Prashant, Julian V. Johnsen, and Katrine V. Løken. "Smoking bans, maternal smoking and birth outcomes." *Journal of Public economics* 115 (2014): 72-93.
- 3. Bharadwaj, Prashant, Juan Pedro Eberhard, and Christopher A. Neilson. "Health at birth, parental investments, and academic outcomes." *Journal of Labor Economics* 36, no. 2 (2018): 349-394.
- 4. Blake, Kevin V., Lyle C. Gurrin, Sharon F. Evans, Lawrence J. Beilin, Louis I. Landau, Fiona J. Stanley, and John P. Newnham. "Maternal cigarette smoking during pregnancy, low birth weight and subsequent blood pressure in early childhood." *Early Human Development* 57, no. 2 (2000): 137-147.
- 5. Butler, N. R., H. Goldstein, and E. M. Ross. "Cigarette smoking in pregnancy: its influence on birth weight and perinatal mortality." *BMJ* 2, no. 5806 (1972): 127-130.
- 6. Carpenter, Christopher, and Philip J. Cook. "Cigarette taxes and youth smoking: new evidence from national, state, and local Youth Risk Behavior Surveys." *Journal of Health Economics* 27, no. 2 (2008): 287-299.
- 7. Centers for Disease Control and Prevention. QuickStats: Number of Deaths from 10 Leading Causes—National Vital Statistics System, United States, 2010. *Morbidity and Mortality Weekly Report* (2013):62(08);155.
- 8. Chaloupka, Frank J., and Kenneth E. Warner. "The economics of smoking." *Handbook of Health Economics* 1 (2000): 1539-1627.
- 9. Chen, Aimin, Michael L. Pennell, Mark A. Klebanoff, Walter J. Rogan, and Matthew P. Longnecker. "Maternal smoking during pregnancy in relation to child overweight: follow-up to age 8 years." *International Journal of Epidemiology* 35, no. 1 (2005): 121-130.
- 10. Chou, Shin-Yi, Michael Grossman, and Henry Saffer. "An economic analysis of adult obesity: results from the Behavioral Risk Factor Surveillance System." *Journal of Health Economics* 23, no. 3 (2004): 565-587.
- 11. Cnattingius, Sven. "The epidemiology of smoking during pregnancy: smoking prevalence, maternal characteristics, and pregnancy outcomes." *Nicotine & tobacco research* 6, no. Suppl_2 (2004): S125-S140.
- 12. Comstock, George W., F. K. Shah, M. B. Meyer, and H. Abbey. "Low birth weight and neonatal mortality rate related to maternal smoking and socioeconomic status." *American Journal of Obstetrics and Gynecology* 111, no. 1 (1971): 53-59.
- 13. Evans, William N., and Jeanne S. Ringel. "Can higher cigarette taxes improve birth outcomes?." *Journal of Public Economics*, 72, no. 1 (1999): 135-154.
- 14. Evans, William N., and Jeanne S. Ringel. "Cigarette taxes and smoking during pregnancy." *American Journal of Public Health* 91, no. 11 (2001): 1851-1856.
- 15. Fried, Peter A., Barbara Watkinson, and Robert Gray. "Growth from birth to early adolescence in offspring prenatally exposed to cigarettes and marijuana." *Neurotoxicology and Teratology*21, no. 5 (1999): 513-525.
- 16. Gilman, Stephen E., Hannah Gardener, and Stephen L. Buka. "Maternal smoking during pregnancy and children's cognitive and physical development: a causal risk factor?" *American Journal of Epidemiology* 168, no. 5 (2008): 522-531.
- 17. Hanrahan, John P., Ira B. Tager, Mark R. Segal, Tor D. Tosteson, Robert G. Castile, Helen Van Vunakis, Scott T. Weiss, and Frank E. Speizer. "The effect of maternal smoking during pregnancy on early infant lung function." *American Review of Respiratory Disease* 145, no.

- 5 (1992): 1129-1135.
- 18. Harris, Jeffrey E., Ana Inés Balsa, and Patricia Triunfo. "Tobacco control campaign in Uruguay: impact on smoking cessation during pregnancy and birth weight." *Journal of Health Economics* 42 (2015): 186-196.
- 19. Howe, Laura D., Alicia Matijasevich, Kate Tilling, Marie-Jo Brion, Sam D. Leary, George Davey Smith, and Debbie A. Lawlor. "Maternal smoking during pregnancy and offspring trajectories of height and adiposity: comparing maternal and paternal associations." *International Journal of Epidemiology* 41, no. 3 (2012): 722-732.
- 20. Huizink, Anja C., and Eduard JH Mulder. "Maternal smoking, drinking or cannabis use during pregnancy and neurobehavioral and cognitive functioning in human offspring." *Neuroscience & Biobehavioral Reviews* 30, no. 1 (2006): 24-41.
- 21. Ino, Toshihiro. "Maternal smoking during pregnancy and offspring obesity: Meta-analysis." *Pediatrics International* 52, no. 1 (2010): 94-99.
- 22. Islam, M. Kamrul, Sherman Folland, and Oddvar M. Kaarbøe. "Social capital and cigarette smoking: New empirics featuring the Norwegian HUNT data." *Economics & Human Biology* 26 (2017): 174-185.
- 23. Kabir, Zubair, V. Clarke, R. Conroy, E. McNamee, S. Daly, and Luke Clancy. "Low birthweight and preterm birth rates 1 year before and after the Irish workplace smoking ban." *BJOG: An International Journal of Obstetrics & Gynaecology* 116, no. 13 (2009): 1782-1787.
- 24. Key, Alexandra PF, Melissa Ferguson, Dennis L. Molfese, Kelley Peach, Casey Lehman, and Victoria J. Molfese. "Smoking during pregnancy affects speech-processing ability in newborn infants." *Environmental Health Perspectives* 115, no. 4 (2007): 623.
- 25. Li, L., H. Peters, A. Gama, M. I. M. Carvalhal, H. G. M. Nogueira, V. Rosado-Marques, and C. Padez. "Maternal smoking in pregnancy association with childhood adiposity and blood pressure." *Pediatric Obesity* 11, no. 3 (2016): 202-209.
- 26. Lien, Diana S., and William N. Evans. "Estimating the impact of large cigarette tax hikes the case of maternal smoking and infant birth weight." *Journal of Human Resources*, 40, no. 2 (2005): 373-392.
- 27. Liu, Feng, Ning Zhang, Kai-Wen Cheng, and Hua Wang. "Reduced smoking and rising obesity: Does smoking ban in the workplace matter?" *Economics Letters* 108, no. 3 (2010): 249-252.
- 28. Maessen, Sarah E., Fredrik Ahlsson, Maria Lundgren, Wayne S. Cutfield, and José GB Derraik. "Maternal smoking early in pregnancy is associated with increased risk of short stature and obesity in adult daughters." *Scientific Reports*, 9, no. 1 (2019): 4290.
- 29. Mendez, Michelle A., Maties Torrent, Carlos Ferrer, Nuria Ribas-Fitó, and Jordi Sunyer. "Maternal smoking very early in pregnancy is related to child overweight at age 5–7 y." *The American Journal of Clinical Nutrition* 87, no. 6 (2008): 1906-1913.
- 30. Meyer, M. B., and J. A. Tonascia. "Maternal smoking, pregnancy complications, and perinatal mortality." *American Journal of Obstetrics & Gynecology* 128, no. 5 (1977): 494-502.
- 31. Milberger, Sharon, Joseph Biederman, Stephen V. Faraone, Lisa Chen, and Janice Jones. "Is maternal smoking during pregnancy a risk factor for attention deficit hyperactivity disorder in children?" *The American Journal of Psychiatry* 153, no. 9 (1996): 1138.
- 32. Miller, Herbert C., Khatab Hassanein, and Paul A. Hensleigh. "Fetal growth retardation in relation to maternal smoking and weight gain in pregnancy." *American Journal of Obstetrics and Gynecology* 125, no. 1 (1976): 55-60.

- 33. Ogden, Cynthia L., Margaret D. Carroll, and Katherine M. Flegal. "High body mass index for age among US children and adolescents, 2003-2006." *JAMA*, 299, no. 20 (2008): 2401-2405.
- 34. Ogden, Cynthia L., Margaret D. Carroll, Lester R. Curtin, Molly M. Lamb, and Katherine M. Flegal. "Prevalence of high body mass index in US children and adolescents, 2007-2008." *JAMA*, 303, no. 3 (2010): 242-249.
- 35. Ogden, Cynthia L., Margaret D. Carroll, Brian K. Kit, and Katherine M. Flegal. "Prevalence of obesity and trends in body mass index among US children and adolescents, 1999-2010." *JAMA*, 307, no. 5 (2012): 483-490.
- 36. Ogden, Cynthia L., Margaret D. Carroll, Hannah G. Lawman, Cheryl D. Fryar, Deanna Kruszon-Moran, Brian K. Kit, and Katherine M. Flegal. "Trends in obesity prevalence among children and adolescents in the United States, 1988-1994 through 2013-2014." *JAMA* 315, no. 21 (2016): 2292-2299.
- 37. Oken, Emily, Susanna Y. Huh, Elsie M. Taveras, Janet W. Rich-Edwards, and Matthew W. Gillman. "Associations of maternal prenatal smoking with child adiposity and blood pressure." *Obesity Research* 13, no. 11 (2005): 2021-2028.
- 38. Ong, Ken KL, Marion L. Ahmed, Pauline M. Emmett, Michael A. Preece, and David B. Dunger. "Association between postnatal catch-up growth and obesity in childhood: prospective cohort study." *BMJ*, 320, no. 7240 (2000): 967-971.
- 39. Orzechowski, W., and R. C. Walker. "The tax burden on tobacco: volume 51, 1970-2016. Washington DC: Centers for Disease Control and Prevention." *National Center for Chronic Disease Prevention and Health Promotion, Office on Smoking and Health* (2017).
- 40. Papke, Leslie E., and Jeffrey M. Wooldridge. "Panel data methods for fractional response variables with an application to test pass rates." *Journal of Econometrics* 145, no. 1-2 (2008): 121-133.
- 41. Permutt, Thomas, and J. Richard Hebel. "Simultaneous-equation estimation in a clinical trial of the effect of smoking on birth weight." *Biometrics* (1989): 619-622.
- 42. Pieroni, Luca, and Luca Salmasi. "The effect of smoking habit changes on body weight: Evidence from the UK." *Economics & Human Biology* 20 (2016): 1-13.
- 43. Restrepo, Brandon J. "Parental investment responses to a low birth weight outcome: who compensates and who reinforces?." *Journal of Population Economics* 29, no. 4 (2016): 969-989.
- 44. Rosenzweig, Mark R., and Kenneth I. Wolpin. "Inequality at birth: The scope for policy intervention." *Journal of Econometrics* 50, no. 1-2 (1991): 205-228.
- 45. Salsberry, Pamela J., and Patricia B. Reagan. "Taking the long view: the prenatal environment and early adolescent overweight." *Research in Nursing & Health* 30, no. 3 (2007): 297-307.
- 46. Sexton, Mary, and J. Richard Hebel. "A clinical trial of change in maternal smoking and its effect on birth weight." *JAMA*, 251, no. 7 (1984): 911-915.
- 47. Simon, David. "Does early life exposure to cigarette smoke permanently harm childhood welfare? Evidence from cigarette tax hikes." *American Economic Journal: Applied Economics*8, no. 4 (2016): 128-59.
- 48. Stick, Stephen M., Paul R. Burton, L. Gurrin, P. D. Sly, and P. N. LeSouef. "Effects of maternal smoking during pregnancy and a family history of asthma on respiratory function in newborn infants." *The Lancet* 348, no. 9034 (1996): 1060-1064.
- 49. Stock, James H., Jonathan H. Wright, and Motohiro Yogo. "A survey of weak instruments and weak identification in generalized method of moments." *Journal of Business & Economic Statistics* 20, no. 4 (2002): 518-529.
- 50. Suzuki, Kohta, Naoki Kondo, Miri Sato, Taichiro Tanaka, Daisuke Ando, and Zentaro Yamagata. "Gender differences in the association between maternal smoking during

- pregnancy and childhood growth trajectories: multilevel analysis." *International Journal of Obesity* 35, no. 1 (2011): 53.
- 51. Thapar, Anita, Tom Fowler, Frances Rice, Jane Scourfield, Marianne van den Bree, Hollie Thomas, Gordon Harold, and Dale Hay. "Maternal smoking during pregnancy and attention deficit hyperactivity disorder symptoms in offspring." *American Journal of Psychiatry* 160, no. 11 (2003): 1985-1989.
- 52. Vik, Torstein, Geir Jacobsen, Lars Vatten, and Leiv S. Bakketeig. "Pre-and post-natal growth in children of women who smoked in pregnancy." *Early Human Development* 45, no. 3 (1996): 245-255.
- 53. Von Kries, Rüdiger, André Michael Toschke, Berthold Koletzko, and William Slikker Jr. "Maternal smoking during pregnancy and childhood obesity." *American Journal of Epidemiology* 156, no. 10 (2002): 954-961.
- 54. Wakschlag, Lauren S., Benjamin B. Lahey, Rolf Loeber, Stephanie M. Green, Rachel A. Gordon, and Bennett L. Leventhal. "Maternal smoking during pregnancy and the risk of conduct disorder in boys." *Archives of General Psychiatry* 54, no. 7 (1997): 670-676.
- 55. Wang, Youfa, and Hyunjung Lim. "The global childhood obesity epidemic and the association between socio-economic status and childhood obesity." (2012): *International Review of Psychiatry*, 24(3): 176-188.
- 56. Ward, Corinne, Sarah Lewis, and Tim Coleman. "Prevalence of maternal smoking and environmental tobacco smoke exposure during pregnancy and impact on birth weight: retrospective study using Millennium Cohort." *BMC Public Health* 7, no. 1 (2007): 81.
- 57. Weitzman, Michael, Steven Gortmaker, and Arthur Sobol. "Maternal smoking and behavior problems of children." *Pediatrics* 90, no. 3 (1992): 342-349.
- 58. Wasserman, Jeffrey, Willard G. Manning, Joseph P. Newhouse, and John D. Winkler. "The effects of excise taxes and regulations on cigarette smoking." *Journal of Health Economics* 10, no. 1 (1991): 43-64.
- 59. Whitaker, Robert C. "Predicting preschooler obesity at birth: the role of maternal obesity in early pregnancy." *Pediatrics*114, no. 1 (2004): e29-e36.
- 60. Wideroe M, Vik T, Jacobsen G, and Bakketeig LS. "Does maternal smoking during pregnancy cause childhood overweight?". *Paediatric and Perinatal Epidemiology*, 17(2), . (2003) 171-179.
- 61. Wooldridge, J. M. (2011, July). Fractional response models with endogeneous explanatory variables and heterogeneity. In CHI11 Stata Conference (No. 12). Stata Users Group.

Appendix

Table A.1

Two-stage least square estimates of relationship between maternal smoking during pregnancy and child bodyweight outcomes with additional set of covariates

	Birthweight	Overweight	Overweight	Underweight	Underweight
		(BMI)	(WFA)	(BMI)	(WFA)
	(1)	(2)	(3)	(4)	(5)
Prenatal smoking	-0.493**	-0.369*	-0.366	-0.055	-0.216
	(0.232)	(0.199)	(0.165)	(0.131)	(0.159)
First stage (Demons	dant vaniahla.	In diagram of r	on on at all and ob	·	
First stage (Depend	ieni variabie:	inaicaior oj p	ргенатат ѕток	ing)	
Tobacco tax	-0.275***	-0.377***	-0.312***	-0.377***	-0.312***
	(0.047)	(0.056)	(0.055)	(0.056)	(0.055)
	•				•
Partial F-stat	10.24	44.48	31.47	44.48	31.47
Observation	6024	4478	4478	4478	4478

Notes: All models control for child characteristics, mother characteristics, and state fixed effects. All models control for child characteristics, mother characteristics, and state fixed effects. Additional characteristics include prenatal doctor visits and vitamin intake during pregnancy. Robust standard errors are clustered at the mother level and are reported in parentheses. ***, **, * = statistically different from zero at the 1%, 5%, 10% level.

Table A.2

OLS and Mother Fixed Effects estimates of relationship between maternal smoking during pregnancy and child weight

Child age	2 years		3 years		4 y	ears	5 years		
	OLS	FE	OLS	FE	OLS	FE	OLS	FE	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	
Prenatal smoking	0.002 (0.199)	-2.565 (3.103)	0.439** (0.173)	0.756 (1.429)	-0.173 (0.152)	0.136 (1.927)	-0.048 (0.210)	0.620 (2.529)	
Observations	683		1571		1459		840		

Notes: All models control for child characteristics, mother characteristics, and state fixed effects. All models control for child characteristics, mother characteristics, and state fixed effects. Robust standard errors are clustered at the mother level and are reported in parentheses. ***, **, * = statistically different from zero at the 1%, 5%, 10% level. We do not estimate 2-SLS models for age-specific effects of maternal smoking during pregnancy as small sample size at each age likely reduces the precision of IV estimates.